Active design of built environments for increasing levels of physical activity in adults: the ENABLE London natural experiment study

Christopher G Owen,^{1*} Elizabeth S Limb,¹ Claire M Nightingale,¹ Alicja R Rudnicka,¹ Bina Ram,¹ Aparna Shankar,¹ Steven Cummins,² Daniel Lewis,² Christelle Clary,² Ashley R Cooper,^{3,4} Angie S Page,^{3,4} Duncan Procter,^{3,4} Anne Ellaway,⁵ Billie Giles-Corti,⁶ Peter H Whincup¹ and Derek G Cook¹

¹Population Health Research Institute, St George's, University of London, London, UK ²Population Health Innovation Lab, London School of Hygiene & Tropical Medicine, London, UK

³Centre for Exercise, Nutrition and Health Sciences, School for Policy Studies, Faculty of Social Sciences and Law, University of Bristol, Bristol, UK

⁴National Institute for Health Research Bristol Biomedical Research Centre,

University Hospitals Bristol NHS Foundation Trust, University of Bristol, Bristol, UK ⁵Medical Research Council and Scottish Government Chief Scientist Office Social and Public Health Sciences Unit, Institute of Health & Wellbeing, University of Glasgow, Glasgow, UK

⁶National Health and Medical Research Council Centre of Research Excellence in Healthy Liveable Communities, Centre for Urban Research, Royal Melbourne Institute of Technology University, Melbourne, VIC, Australia

*Corresponding author cowen@sgul.ac.uk

Declared competing interests of authors: Christopher G Owen reports that he is an active member of the Public Health Research (PHR) Research Funding Committee (16 June 2015 to present). Steven Cummins was a member of the PHR Research Funding Board (13 May 2009 to 12 May 2015). Diabetes mellitus and obesity prevention research at St George's, University of London is supported by the National Institute for Health Research (NIHR) Collaboration for Leadership in Applied Health Research and Care (CLAHRC) South London. Bina Ram was supported by a St George's, University of London, PhD Studentship. Anne Ellaway is funded by the UK Medical Research Council as part of the Neighbourhoods and Communities Programme (MC_UU_12017-10). Billie Giles-Corti is supported by a National Health and Medical Research Council Principal Research Fellowship (#1107672). Ashley R Cooper and Angie S Page are supported by the NIHR Biomedical Research Centre at the University Hospitals Bristol NHS Foundation Trust and the University of Bristol.

Published September 2020 DOI: 10.3310/phr08120

Plain English summary

The ENABLE London natural experiment study Public Health Research 2020; Vol. 8: No. 12 DOI: 10.3310/phr08120

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

What was the problem?

Physical activity levels are too low, and population-based strategies to improve physical activity levels have had modest effects, at best. However, small improvements across communities could have significant effects on health outcomes at a population level. Changes in the neighbourhood built environment may offer an attractive way to increase physical activity levels, given their potential reach.

What did we do?

East Village, the former London 2012 Olympic and Paralympic Games Athletes' Village, was repurposed to provide social, affordable (intermediate) and market-rent housing with high levels of walkability and close proximity to improved public transport. The Examining Neighbourhood Activities in Built Living Environments in London (ENABLE London) study was a natural experiment that aimed to examine whether or not objectively measured physical activity levels, body fat and mental health/well-being show sustained change after 2 years in individuals relocating to East Village compared with a control population living outside East Village throughout.

What did we find?

East Village resulted in a marked improvement in objective measures of the built environment (including improved walkability and access to public transport) and more positive perceptions of lower crime (i.e. less vandalism, feeling safer to walk in the neighbourhood and less of a presence of threatening groups) and quality of the built environment among residents (particularly among those moving to East Village social housing). Despite these sizeable improvements, there were only modest increases in physical activity levels (particularly in the number of daily steps) and few differences in higher levels of physical activity (i.e. moderate to vigorous physical activity), sedentary time, measures of body fatness and other health-related outcomes.

What does this mean?

This study, using a robust design, showed modest improvements in physical activity and other health-related behaviours associated with moving to accommodation designed for active living, suggesting that improving the built environment alone is insufficient to increase population levels of physical activity. Further work is needed to examine how changes in the built environment can lead to improved health behaviours, such as physical activity.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Owen *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Public Health Research

ISSN 2050-4381 (Print)

ISSN 2050-439X (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PHR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/phr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Public Health Research journal

Reports are published in *Public Health Research* (PHR) if (1) they have resulted from work for the PHR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Public Health Research* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

PHR programme

The Public Health Research (PHR) programme, part of the National Institute for Health Research (NIHR), is the leading UK funder of public health research, evaluating public health interventions, providing new knowledge on the benefits, costs, acceptability and wider impacts of non-NHS interventions intended to improve the health of the public and reduce inequalities in health. The scope of the programme is multi-disciplinary and broad, covering a range of interventions that improve public health.

For more information about the PHR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/ public-health-research.htm

This report

The research reported in this issue of the journal was funded by the PHR programme as project number 12/211/69. The contractual start date was in June 2014. The final report began editorial review in February 2019 and was accepted for publication in June 2019. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PHR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the PHR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PHR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Owen *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

Editor-in-Chief of Public Health Research and NIHR Journals Library

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk